

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Rünger, Silke

Working Paper The effect of Germany's Tax Reform Act 2001 on corporate ownership: Insights from disposals of minority blocks

arqus Discussion Paper, No. 114

Provided in Cooperation with: argus - Working Group in Quantitative Tax Research

Suggested Citation: Rünger, Silke (2011) : The effect of Germany's Tax Reform Act 2001 on corporate ownership: Insights from disposals of minority blocks, arqus Discussion Paper, No. 114, Arbeitskreis Quantitative Steuerlehre (arqus), Berlin

This Version is available at: https://hdl.handle.net/10419/45017

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

arqus

Arbeitskreis Quantitative Steuerlehre

www.arqus.info



Diskussionsbeitrag Nr. 114

Silke Rünger

The Effect of Germany's Tax Reform Act 2001 on Corporate Ownership – Insights from Disposals of Minority Blocks

Januar 2011

arqus Diskussionsbeiträge zur Quantitativen Steuerlehre arqus Discussion Papers in Quantitative Tax Research ISSN 1861-8944

The Effect of Germany's Tax Reform Act 2001 on Corporate Ownership - Insights from Disposals of Minority Blocks

Silke Rünger^{*}

Abstract: The German tax reform act 2001 changed the corporate tax system from a full imputation system to a half income system. Along with this change, the taxation of equity investments changed as well. Using data on 459 disposals of minority blocks over the period 1997-2006, this paper analyzes the effect of TRA 2001 on the demand for corporate shares of different owner types and on corporate ownership concentration. We show that TRA 2001 was able to fulfill government's expectations about an increase in blocks bought by individual owners. With respect to ownership concentration, we find tax incentives not to be strong enough to lead to a reduction in overall concentration of corporate ownership.

Keywords: corporate ownership, marginal tax rate on equity, minority blocks, Germany

JEL classification: G11, G34, H24, H32

^{*} The author would like to thank Rainer Niemann and participants at the Workshop for Empirical Research in Corporate Taxation in Mannheim and the CESifo Conference on Corporate Taxes and Corporate Governance in Munich for helpful comments. Financial support by the Austrian Science Fund (FWF) [P 22324-G1] is gratefully acknowledged.

1 Introduction

Promoted as the 'best Christmas gift in the history of the German stock index'¹ the German tax reform act (TRA) 2001 was believed to substantially alter the system of corporate control and corporate ownership in Germany. It changed the corporate tax system from a full imputation system to a half-income system, a classical corporate tax system with shareholder relief elements. Along with the change in the tax system, the taxation of equity investments for corporate and individual investors changed as well.

The taxation of equity investments can lead to two distortional effects, an allocation (clientele) effect and a timing effect. Whereas the first one distorts the investment decision itself, the second one distorts the decision about when to sell an existing investment. According to the allocation effect the taxation of equity influences the allocation of a shareholder's portfolio. Corporations can distribute their profits to shareholders either via dividends or share repurchases. The taxation of these two options is usually different; dividends are taxed on an annual basis, whereas share repurchases resulting in capital gains are taxed upon realization. Depending on the tax rates imposed, shareholders will either prefer assets with higher returns that are taxed upon realization or assets with higher dividend payments that are taxed on an annual basis. Changes in shareholder taxation will alter the allocation of a shareholder's portfolio and lead to a change in corporate ownership.²

By repealing the corporate capital gains tax, the German government was expecting an increase in disposals of corporate holdings leading to substantial changes in corporate ownership. Along with that, a reduction in the network of German corporations, connected mostly through minority blocks, was expected.

Whereas several papers have already analyzed the effect of TRA 2001 on single corporate divestiture choices³, to the best of our knowledge, no paper has so far dealt with general ownership effects of TRA 2001. Our analysis differs from prior studies not only by evaluating the effects of TRA 2001 on the demand for corporate shares, but also by the fact that we can use a unique data set, where information about the seller and the acquirer of an owner block can be directly observed. In addition, our sample is not limited to listed corporations, but also includes investments in non-listed corporations, that form up the majority of German corporations.

¹ See Schürmann (2005), p. 84.

² See Desai / Gentry (2004).

³ See e.g. Gieralka / Drajewicz (2001), Edwards / Lang / Maydew / Shackelford (2004) and Watrin / Benhof (2008).

We are able to show that TRA 2001 was able to fulfill government's expectations about an increase in blocks bought by individual owners. It can also be shown, that corporate and individual owners have different preferences with respect to firm specific characteristics. Corporate owners prefer to hold blocks in non listed companies and to acquire blocks sold by non financial companies, whereas individual owners are more likely to hold blocks in listed companies and acquire blocks sold by financial companies. With respect to ownership concentration, we find tax incentives not to be strong enough to lead to a reduction in overall concentration of corporate ownership.

The remainder of this paper is organized as follows: Chapter 2 gives an overview of the legal background as well as prior research dealing with corporate ownership. Chapter 3 presents the investment model used in the paper, details about measurement issues related to ownership concentration and the research hypotheses. Our sample and the regression models used for archival analysis are carried out in chapter 4. Chapter 5 presents the results of our analysis, chapter 6 concludes.

2 Institutional Background and Prior Research

2.1 Institutional Background

By the beginning of the new century, the German network-orientated corporate governance model was characterized by the predominance of large shareholders and by the fact that large German banks acted as important shareholders in many industrial companies.⁴ A term often used to describe this situation is 'Germany Inc.' (Deutschland AG). It refers to a network of German listed companies, especially the financial service providers Allianz, Commerzbank, Deutsche Bank, Dresdner Bank, Hypo-Vereinsbank and Munich Re, which was closely connected through board relations and capital cross-holdings.

Several studies have so far analyzed the distribution of control among German firms. Becht / Böhmer (1999) examine the distribution of voting blocks among 430 listed German corporations. Their results show a mean of the largest voting block of 58.9%; the second and third largest block do not add much voting power (9.8% taken together). Only a fifth of all German corporations has more than two owners. Voting blocks are found to be concentrated around the important control thresholds of 25%, 50% and 75%. Not surprisingly, German banks and insurance companies show the highest number of blocks

 $[\]overline{4}$ See Shleifer / Vishny (1997).

held (Deutsche Bank 27 blocks, Allianz 25 blocks, Dresdner Bank and Munich Re 14 blocks each).⁵ In an international survey, Becht / Röell (1999) compare ownership structures of listed corporations in eight European countries and the United States. They show that voting blocks are much larger in continental Europe compared to the United Kingdom and the United States. For 374 listed German corporations they find the largest block to be 52.1% on average. Only Italian corporations show a higher concentration, the largest voting block being 54.53%. For the UK sample the largest voting block accounted for only 9.9% on average and in the United States it was below the 5% disclosure threshold.⁶

The intense network of German cross-holdings and the high concentration of ownership was believed to cause several disadvantages for German firms. First of all, crossholdings lead to the fact, that control is limited and control mechanisms are only executed among the corporations in the network. For example, under German commercial law, a bank depositing shares for its clients is able to execute proxy voting rights in shareholders' meetings.⁷ Due to this fact, in 1986 at the annual general meeting of Deutsche Bank the company itself accounted for nearly 48% of all present voting rights. It seems not surprising, that the high ownership concentration of German corporations was seen as a major obstacle for the development of the German capital market and even called a locational disadvantage for Germany.⁸

Growing international competition along with alternative ways of financing for industrial companies forced German banks to concentrate on their core competencies and rebalance their portfolios of industrial minority holdings. The fact that TRA 2001 completely repealed the taxation of corporate capital gains offered the possibility to realize a large amount of hidden reserves without diminishing shareholder value.⁹ The disposal of corporate holdings was expected to increase the proportion of shares traded and thus the free float of securities.¹⁰ Thus, an increase in blocks bought by individual investors along with a reduction in ownership concentration was expected.

⁵ See Becht / Böhmer (1999), p. 37f.

⁶ See Becht / Röell (1999), p. 1052.

⁷ Although voting instructions must be sought by law, studies have shown that only 2-3% of all shareholders give instructions about how to vote. See Becht / Böhmer (1999), p. 13.

⁸ See Adams (1994), p. 152ff.

⁹ See Beyer (2003), p. 122, Breuer (1998), p. 538, Frick / Lehmann (2004), p. 123 and Wieandt / Smith (2006), p. 14.

 $^{^{10}\,}$ See $\,$ Cioffi (2002), p. 379.

2.2 Legal Background: The German Tax Reform Act 2001

The taxation of an equity investment is determined by the taxation on the level of the corporation and on the level of the shareholder. Corporations can distribute their profits to shareholders either via dividends or share repurchases, resulting in a capital gain for the investor. TRA 2001 changed the taxation on the corporate as well as on the shareholder level, including changes in the taxation of dividends as well as in the taxation of capital gains.

On the corporate level, the tax system was changed from a full imputation system to a half-income system, a classical corporate tax system with shareholder relief elements. In addition, the corporate tax rate τ_c was reduced from 40% for retained earnings and 30% for distributed earnings to a uniform rate of 25%.¹¹

Under the full imputation system, dividends were taxed on the shareholder level at the dividend tax rate τ_d^{IMP} and corporate taxes paid were credited against the tax liability of the shareholder in order to avoid double taxation. If the shareholder was a German corporation (individual), dividends received were effectively taxed at the corporate (individual) tax rate τ_c^{IMP} (τ_i^{IMP}), due to the full imputation granted.

With the change from the full imputation system to the half-income system, dividends were taxed on the corporate level at the uniform tax rate τ_c^{HI} . In order to avoid double taxation of corporate earnings, intercorporate dividends became tax exempt on the shareholder level. Still, 5% of dividends received were classified as non-deductible expenditures, resulting in an overall tax burden for intercorporate dividends of $1.05 \cdot \tau_c^{HI}$. As a shareholder relief element for individual shareholders, only half of the dividends received were taxed at the ordinary income tax rate τ_i^{HI} . The tax rate for dividends received by individual investors was therefore determined as follows:

$$\tau_{d(indiv)}^{HI} = \tau_c^{HI} + (1 - \tau_c^{HI}) \cdot \tau_i^{HI} \cdot 0.5 \tag{1}$$

Under the full imputation system, capital gains realized by corporate shareholders were fully taxable, thus $\tau_{g(cor)}^{IMP} = \tau_c^{IMP}$. Because of the government's aim to tax dividends and capital gains at the same tax rate, corporate capital gains realized on the disposal of shares in domestic and foreign corporations became tax-exempt under the half-income

¹¹ Throughout this paper, we do not consider the German solidary surcharge. It accounts for 5.5% (1998: 7.5%) of the tax burden for corporations as well as individuals.

system.¹²

Capital gains realized by German individual investors were only taxable if an investment was classified as a substantial interest, defined as an investment above a certain threshold. TRA 2001 reduced this threshold from 10% to 1%. Under the full imputation system, capital gains realized on the disposal of substantial holdings were taxed at a reduced income tax rate, thus $\tau_{g(indiv)}^{IMP} = \tau_{i(red)}^{IMP}$. Under the half-income system, capital gains received by individual shareholders were aimed to be taxed at the same tax rate as dividends, thus $\tau_{d(indiv)}^{HI} = \tau_{g(indiv)}^{IMP}$.

Two recent studies have empirically examined changes in corporate ownership of German firms. Kengelbach / Roos (2006) analyze the network of German corporate holdings over the period 1996-2004. According to their hypotheses a reduction in ownership concentration can be observed, if the amount of German subsidiary companies held by German parent companies or the amount of German companies with a majority blockholder or the mean of the largest voting block of German corporations or the amount of corporate crossholdings decreases over time.¹³ Analyzing the ownership structures of German listed corporations, the authors find empirical evidence for all hypotheses.

Weber (2009) measures ownership concentration of German listed corporations by summing up the three largest voting blocks over the period 1999-2005. Besides a decrease in ownership concentration she finds the number of blocks held by individuals owners to rise, although the mean size of their blocks is falling.¹⁴

2.3 Prior Empirical Research

Prior empirical research has shown, that the ownership structure of a firm is endogenously determined by firm-specific characteristics and the firm's competitive environment.¹⁵ In their seminal paper, Demsetz / Lehn (1985) empirically analyze the determinants of corporate ownership structures and find the following four firm-specific factors to system-atically influence ownership concentration:

• Size: The larger a firm, the greater its market value and therefore the higher the price, an investor has to pay for a given fraction of the firm. In addition, a given

 $^{^{12}\,}$ Again, 5% of capital gains realized were classified as non-deductible expenditures for tax purposes.

 $^{^{13}}$ See Kengelbach / Roos (2006), p. 15.

¹⁴ See Weber (2009), p. 60ff.

¹⁵ See Short (1994).

degree of control requires a smaller share of the firm. Both these effects imply lower ownership concentration for larger firms.

- Control Potential: Control potential is defined as the wealth gain achievable through more effective monitoring of managerial performance by a firm's owner. Managerial behavior is more difficult to monitor if a firm operates in a less predictable environment. The noisier a firm's environment, the more concentrated ownership structures can be expected.
- **Regulation:** Regulation within an industry can be seen as a substitute for managerial control by owners. Ownership structures are therefore expected to be less concentrated in regulated industries.
- Amenity Potential: Shareholders who own a large fraction of shares of a firm are in the position to control for the management serving their interests. Despite the fact that shareholders want a firm to maximize its profit, they also might demand nonpecuniary income associated with firm specific characteristics. For firms with higher amenity potential, ownership structures are expected to be more concentrated.

Testing these factors for a sample of 511 listed US corporations shows that all variables have a significant influence and all coefficients have the expected sign. Based on the results of Demsetz / Lehn (1985), several other studies testing the determinants of ownership concentration for different countries have been published.¹⁶ The variables used in these papers generally remain the four variables used by Demsetz / Lehn (1985). The only additional firm-specific factor used in several papers is firm age. Ownership concentration in younger firms is expected to be higher than in old firms since control is passed on from the founders to future shareholders.

Table 1 gives an overview of papers analyzing firm-specific determinants of corporate ownership for various countries. It shows the main variables used in the paper as well as whether these variables were found to have a significant influence on ownership concentration.

{Insert table 1 about here.}

¹⁶ For details concerning these studies see Bergström / Rydquist (1990), Leech / Leahy (1991), Prowse (1992), Bohren / Odegaard (2001), Mak / Li (2001), Lamba / Stapledon (2001), Wojcik (2002), Pindado / de la Torre (2006), Madsen / Smith / Dilling-Hansen (2007) and Rogers / Borges / de Sousa Ribiero / de Sousa (2008).

Contrary to the approach of Demsetz / Lehn (1985) several papers contribute differences in corporate ownership structures to the legal framework of a country. These papers are either based on a cross-country setting or analyze legal changes in a single country.¹⁷ La Porta / Lopez de Silanes / Shleifer / Vishny (1998) calculate corporate ownership concentration for 48 countries by summing up the three largest voting blocks. The results show that ownership concentration is a reaction to poor legal protection of shareholders.

Only a few papers have analyzed the influence of taxes on corporate ownership:

Dahlquist / Robertsson (2000) show, that taxes play an important role in determining foreign ownership in Swedish firms. According to their results foreign investors prefer large companies and companies paying low dividends. The latter effect is interpreted to be driven by the tax advantage of capital gains compared to dividend payments for foreign investors in Sweden.

The evolution of ownership concentration in the US is analyzed by Desai / Dharmapala / Fung (2007). The authors see taxes as a major determinant of corporate ownership structures, i.e. the increase in the progressivity of tax schedules. Time-series analysis starting in 1916 show that the increase in progressivity of the tax scheme has lead to greater diffusion of ownership.

La Porta / Lopez de Silanes / Shleifer (1999) analyze 27 the 20 largest and 10 smallest companies with a market capitalization of common equity of at least 500 million of 27 countries. Instead of measuring ownership concentration, the authors look for the presence of an ultimate owner, a controlling shareholder whose direct and indirect voting rights exceed 20%, and then control for factors influencing the presence of the ultimate owner. In the study two tax variables are taken into account, controlling for whether corporate dividends are taxed and whether consolidated accounting is permitted for tax purposes. The results show that both variables have no significant influence on the presence of an ultimate owner.

Holderness (2009) mentions two limitations regarding empirical ownership structure research. On the one hand, many international surveys use country averages instead of ownership concentration calculated with firm-level data. In addition, small samples of large, listed firms are used. This is problematic since it has been shown that ownership concentration is inversely related to firm size.

 ¹⁷ See La Porta / Lopez de Silanes / Shleifer / Vishny (1998), La Porta / Lopez de Silanes / Shleifer (1999), Beny (2005), La Porta / Lopez de Silanes / Shleifer (2006), Roe (2006) or Djankov / La Porta / Lopez de Silanes / Shleifer (2008).

In this paper we are able to analyze the influence of a change in the tax system on corporate ownership using firm level data and a sample of listed and non-listed firms with no need of aggregation.

3 Model and Research Hypotheses

Due to the tax exemption for corporate capital gains, TRA 2001 was expected to increase the number of disposals of corporate shares and therefore lead to a higher supply of corporate shares starting in 2002.¹⁸ On the other side, as has been shown in chapter 2.2, the taxation of shareholders, thus the taxation of the demand side for corporate shares, has changed as well. In this chapter we are going to show which tax induced demand-side effects were caused by TRA 2001 and why these effects are expected to influence corporate ownership in Germany.

3.1 The Investor's Marginal Tax Rate on Equity

The model presented in this chapter is based on the growth model introduced by Gordon / Shapiro (1956) and Gordon (1963), enriched by tax aspects by Gordon / MacKie-Mason (1990).¹⁹

We assume that an investor holding an equity investment receives a pre tax return of 1, that can be either generated by a dividend payment, a capital gain realized or a mixture of both. Integrating taxation, we assume that corporate profits are taxed at the corporate tax rate τ_c , dividends are taxed at the shareholder's dividend tax rate τ_d and capital gains are taxed at the shareholder's capital gains tax rate τ_g .²⁰ Depending on the tax system, corporate taxes may or may not be imputed. In general, the investors's marginal tax rate on equity is given by:

$$\tau_e = 1 - \left[(1 - \tau_c) \cdot \left(d \cdot (1 - \tau_d) + (1 - d) \cdot (1 - \tau_g) \right) \right]$$
(2)

¹⁸ For an analytical explanation see Edwards / Lang / Maydew / Shackelford (2004) or Watrin / Benhof (2008).

¹⁹ For a general analysis of shareholder level taxation on investment decisions with different corporate tax regimes, see Sureth / Langeleh (2007).

²⁰ Papers often take into account the fact that capital gains are taxed only upon realization and taxation can be deferred until the end of the investment. Usually a factor α is integrated into the model to reduce the statutory tax rate for capital gains to an effective tax rate. Since this effect is the same for all type of owners and both tax regimes, we do not integrate it into the model for simplification.

Under the German full imputation system, taxes implied on the corporate level were fully imputed on the shareholder level. As a consequence, dividends were effectively taxed at the shareholder's marginal dividend tax rate. The investor's marginal tax rate on equity under the full imputation system is therefore given by

$$\tau_e^{IMP} = 1 - \left[\left(d \cdot (1 - \tau_d^{IMP}) \right) + \theta \cdot \left(\left(1 - \tau_c^{IMP} \right) \cdot (1 - d) \cdot (1 - \tau_g^{IMP}) \right) \right]$$
(3)

with

 $\theta = 1$ for corporate shareholders and individual shareholders > 10% and

 $\theta = 0$ for individual shareholders $\leq 10\%$.

Under the half-income system, taxes paid on the corporate level were no longer imputed on the shareholder level. As a shareholder relief element, dividends and capital gains became tax exempt for corporate investors and only half of the dividends and capital gains were taxable for individual investors (see chapter 2.2).

The investor's marginal tax rate on equity under the half income system is therefore given by

$$\tau_e^{HI} = 1 - \left[(1 - \tau_c^{HI}) \cdot (d \cdot (1 - \tau_d^{HI}) + \theta \cdot ((1 - d) \cdot (1 - \tau_g^{HI}))) \right]$$
(4)

with

 $\theta = 1$ for corporate shareholders and individual shareholders > 1% and

 $\theta = 0$ for individual shareholders $\leq 1\%$.

Comparing equations 3 and 4 we can see, that the investor's marginal tax rate on equity depends on three aspects: taxes imposed on the corporate and shareholder level, the dividend payout ratio of the firm and the value of θ indicating whether capital gains are taxable or not.

In order to quantify the effects of TRA 2001 on the demand for corporate shares of single owner types, we calculate marginal tax rates on equity for corporate and individual owners and different dividend ratios. For the calculations we assume that corporate owners represent the ultimate owner for tax purposes. This corresponds with the view that corporate investors do not include the tax burden of their owners into the investment decision. For individual owners we only use the top statutory tax rate for the calculations.

This is due to the fact that several papers have shown that individuals owning corporate shares usually belong to the highest tax bracket, see Jacob (2010). Table 2 shows the results for the marginal tax rates on equity.

{Insert table 2 about here.}

As we can see from table 2, TRA 2001 lowered the marginal tax rate on equity for nearly all types of owners. Due to the lowering of the threshold for non-substantial interest from 10% to 1%, capital gains of individual owners within this bracket became fully taxable starting in 2002. As a result, this group faces a higher marginal tax rate on equity under the half income system than under the full imputation system for companies with dividend ratios below 25%. Additionally, we can see that for individual owners there is a clear tax incentive to hold blocks below the 10% (1%) threshold under the full imputation system (the half-income system).

With respect to different dividend payout ratios we find the marginal tax rate on equity to crucially depend on the dividend payout ratio of the firm under the full imputation system. For corporate owners and individual owners above the 10% threshold there is a clear tax clientele incentive to hold shares with high dividend payout ratios, whereas individual owners below the 10% threshold prefer shares with low dividend ratios. Contrary, the firm's payout policy is of minor importance under the half income system, since marginal tax rates for corporate owners and individual owners above the 1% threshold are unaffected by the firm's dividend payout ratio. Only for individual owners below the 1% threshold there is a tax clientele incentive to hold shares with low dividend ratios. This incentive is stronger than under the full imputation system, since marginal tax rates on equity are more sensitive with respect to d.

3.2 Measurement Issues

In order to be able to measure tax induced effects of TRA 2001 on corporate ownership, we look at two different distortional aspects.

On the one hand we quantify the effects of TRA 2001 on the demand for corporate shares by analyzing whether the number of blocks bought by different owner types has changed due to TRA 2001. On the other hand we measure the influence of TRA 2001 on general ownership concentration. These two aspects do not necessarily have to correspond. Consider e.g. the disposal of an owner block of 10%, which is sold by a corporate owner and acquired by an individual owner. In this case ownership concentration has not changed

since the size of the block remains unchanged. Still, the owner type has changed, since the proportion of blocks held by individual owners has increased. Given this difference, we have to analyze both aspects of corporate ownership changes separately in order to fully evaluate tax induced effects of TRA 2001 on corporate ownership.

Prior literature has measured ownership concentration with different concentration measures. The most commonly used concentration measure is a Herfindahl index, H, which is calculated by summing up the square of single shareholder blocks, a_i .

$$H = \sum_{i=1}^{N} a_i^2 \tag{5}$$

According to this definition the value of the Herfindahl index for a given corporation varies between $\frac{1}{N}$ and 1. The minimum value is achieved if blocks are equally distributed among shareholders and the maximum value is achieved if there is only one shareholder holding 100% of the company's shares. We will use the Herfindahl index as the concentration measure for an analysis of changes in ownership concentration.

3.3 Hypotheses Development

As we have seen in chapter 3.1, TRA 2001 lowered the marginal tax rates on equity for most types of owners. We expect the demand for corporate shares to rise, if marginal tax rates of owners are lower and therefore define the following research hypotheses:

Hypothesis 1a: The number of blocks bought by corporate owners is going to increase due to TRA 2001.

Hypothesis 1b: The number of blocks below the 1% threshold bought by individual owners is going to increase due to TRA 2001.

Hypothesis 1c: The number of blocks above the 10% threshold bought by individual owners is going to increase due to TRA 2001.

We have also seen, that the firm's payout policy has an important influence on the marginal tax rate on equity and therefore define the following research hypotheses with respect to the firm's dividend payout ratio d:

Hypothesis 2a: The number of blocks bought by corporate owners under the full imputation system is going to increase, the higher d.

Hypothesis 2c: The number of blocks above the 10% threshold bought by individual owners under the full imputation system is going to increase, the higher *d*.

Hypothesis 2b: The number of blocks below the 10% threshold bought by individual owners under the full imputation system is going to decrease, the higher d.

Hypothesis 2d: The number of blocks below the 1% threshold bought by individual owners under the half income system is going to decrease, the higher d.

As we have shown in chapter 3.2, we measure ownership concentration using the Herfindahl index. According to the Herfindahl approach ownership concentration decreases if a tax reform includes rules that lower the size of blocks demanded due to a lower tax rate on smaller blocks. TRA 2001 lowered the threshold for substantial interest from 10% to 1% for individual owners and therefore included a tax incentive for individual owners to demand smaller blocks. This also corresponds to the findings of Weber (2009) that the number of blocks held by individuals is rising after TRA 2001, although the mean size of the blocks is falling (see chapter 2.2). We therefore define the following research hypothesis concerning ownership concentration:

Hypothesis 3a: Ownership concentration is going to decrease due to TRA 2001.

We also define a general hypothesis concerning the relationship between an investor's marginal tax rate on equity and corporate ownership concentration. In line with the argumentation of La Porta / Lopez de Silanes / Shleifer (1999), we expect lower marginal tax rates on equity to increase the demand for equity investments and to raise the amount of owners on the market. As a result, the size of the block held by a single owner decreases and ownership concentration decreases as well.

Hypothesis 3b: Ownership concentration is going to decrease, the lower τ_e .

4 Research Design

4.1 Sample of Disposals

In order to test the hypotheses stated in chapter 3.3, we make use of the fact that German corporations have to disclose their ownership structure in their financial statements. Among the databases that provide information about ownership structures, we have chosen Amadeus and Osiris from Bureau van Dijk for this paper. These databases offer a very detailed ownership module listing the name of the owner, the percentage of direct ownership and the country of the owner. In addition, both databases provide detailed company information including financial statement data.

We start the selection process in 1997, the first year detailed ownership information is available from the databases. In order to see which blocks have been sold, we track all available ownership blocks below the 25% threshold²¹ held by German corporations. We end the search in the year 2006 and find 459 disposals of minority blocks during thesample period.²² In order to account for different effects of the corporate tax regimes during the observation period, we define three tax portfolios. By doing so, we are able to distinguish between blocks where both, the acquisition and the disposal took place under the full imputation system (portfolio 1) or under the half income system (portfolio 3). In addition, we are able to separate the effects for blocks that have been acquired under the full imputation system, but sold under the half income system (portfolio 2). We refer to portfolio 2 as the short-term and to portfolio 3 as the long term effects of TRA 2001.

Table 3 shows the distribution of the sample disposals among the three tax portfolios. In addition, we distinguish between blocks sold by financial and non financial companies.²³

{Insert table 3 about here.}

The highest number of disposals is found for portfolio 2, especially for blocks sold by financial companies. One reason why the amount of disposals belonging to portfolio 1 is rather low might be due to the fact, that TRA 2001 was first announced by the end of 1999. Therefore, companies might have deferred the disposal of blocks until the law became effective in 2002 in order to avoid capital gains taxation. This might also be an additional explanation for the majority of blocks sold by financial selling companies belonging to portfolio 2.

In addition to the definition of tax portfolios, we classify the disposals according to the type of the acquirer. We define four owner types: type 1 (German corporations), type 2 (German individuals), type 3 (foreigners) and type 4 (state). In addition, we split

²¹ This level represents the blocking minority in Germany. According to the German Companies Act (AktG) anyone owning more than 25% in another company has substantial control rights, i.e. the right to block changes in the articles.

²² If a company was liquidated during the year of the disposal, we do not include this disposal in the sample in order to avoid a liquidation bias.

²³ This is due to the fact that financial companies were believed to hold a high amount of minority blocks available for sale and were expected to benefit at most from the tax law change, see chapter 2. In addition, the acquisition and disposal of corporate shares is part of the core business for financial companies, but not for non financial companies.

owner type 2 into three further subgroups: type 2a (German individuals $\leq 1\%$), type 2b (German individuals > 1% and $\leq 10\%$) and type 2c (German individuals > 10%).

Both of the databases list owners holding blocks above the threshold of 0.1% individually, whereas owners below this threshold are summed up and disclosed as free float. We therefore categorize free float as small owner blocks held by German individual owners $\leq 1\%$ (owner type 2a).

4.2 Regression Models

We are going to run several regression models for the dynamic evaluation of the change in the demand for corporate shares of single owner types and the change in concentration of corporate ownership due to TRA 2001.

For the evaluation of the effect of TRA 2001 on the demand for corporate shares of single owner types, we run three different specifications of a probit model using the following dichotomous dependant variables:

 $Za_{it} = 0$, if owner type = a $Za_{it} = 1$ otherwise

with a equal to owner type 1, 2a and 2c (see chapter 4.1.

The regression model reads as follows:

$$Za_{it} = \alpha + \beta_1 \cdot PF2_{it} + \beta_2 \cdot PF3_{it} + \beta_3 \cdot d_{it} + \beta_4 \cdot d_{it}PF2_{it} + \beta_5 \cdot d_{it}PF3_{it} + \beta \cdot X'_{it} + \varepsilon_{it}$$
(6)

where Za_{it} is defined as shown before, $PF2_{it}$ and $PF3_{it}$ refer to the tax portfolios defined in chapter 4.1 and d_{it} is the percentage of the firm's profit distributed as dividends. As we have shown in chapter 3.1, we expect different effects for the dividend payout ratio under the full imputation and under the half income system. We therefore integrate the two interactive variables $d_{it}PF2_{it}$ and $d_{it}PF3_{it}$ in the regression model. X'_{it} is a vector of non-tax control variables.

Since not all of the companies in the sample are listed, we have to proxy the dividend ratio d_{it} of non-listed companies as a relationship of the company's profit after taxes $Prof_{it}$ and the change in the shareholder's funds of the company SF_{it} as follows:²⁴

 $^{^{24}}$ Note that we assume that the dividend ratio is 0 if the companies realizes a loss after taxation. In addition, we assume constant dividend ratios over time.

$$d_{it} = \frac{Prof_{it} - (SF_{it} - SF_{it-1})}{Prof_{it}} \tag{7}$$

For all owner types analyzed we expect the number of blocks acquired after TRA 2001 to increase and therefore a positive coefficient for $PF2_{it}$ and $PF3_{it}$.

In addition, we expect a positive coefficient for d_{it} and a negative coefficient for $d_{it}PF2_{it}$ and $d_{it}PF3_{it}$ for corporate investors (owner type 1) and for individual owners > 10% (owner type 2c). For individual owners $\leq 1\%$ (owner type 2a) we expect negative coefficients for d_{it} , $d_{it}PF2_{it}$ and $d_{it}PF3_{it}$.

The firm-specific control variables are taken from corporate ownership research shown in chapter 2.3.²⁵ The size of the company, $Size_{it}$ is measured as the value of total assets in the year of the disposal. Control potential, $lnContPot_{it}$, of the owners is measured as the estimated standard deviation of the company's profit over the three years prior to the disposal. Regulation, Reg_{it} , is a dummy variable taking the value 1 if the company belongs to the financial or utility industry (NACE 1.1. 2-digit codes 40, 41, 65 or 66). The age of the company, Age_{it} is measured in years at the time of the disposal. In addition, we control for differences between listed and non-listed companies, $Listed_{it}$, as well as for differences for the seller being a (non)financial company, NF_{it} , using dummy variables. Both of these controls refer to liquidity constraints indicating that for a listed company (a financial selling company), it will be easier to attract individual owners than for non listed companies (non financial selling companies).

We also control for macroeconomic factors that are expected to influence the demand for corporate shares. Melicher / Ledolter / D'Antonio (1983) have shown a negative correlation between the interest rate and aggregated M&A activity, since higher interest rates increase refinancing costs and thus decrease the demand of potential investors. In addition, the authors have shown that increasing stock prices are positively correlated with M&A activities. Since many other macroeconomic factors such as gross national income, market liquidity or yield curves are highly correlated with interest rates and (or) stock prices, we do not include further macroeconomic control variables. We measure the interest rate, $Interest_t$ as the German interbank 12 month offered rate. Stock prices, $Stock_t$, are measured as the yearly performance of the German stock index DAX.

²⁵ Note that we do not include the amenity potential of the firm in the analysis. This is due to the fact that none of the sample firms belongs to the industries (media, sports) defined to provide amenity potential to owners.

For the evaluation of the effect of TRA 2001 on ownership concentration we calculate a Herfindahl index for the sample firms before and after the disposal. One problem with the Herfindahl index is that information about the size of every single owner block is necessary. The databases usually provide this information, except in the case of free float. As has been mentioned in chapter 4.1, free float is defined as small blocks below the 0.1% threshold held by German individual investors. Thus, if a company's free float accounts for 5% this does not correspond to a single owner block accounting for 5%. Excluding free float blocks from the calculation and calculating concentration only for the remaining owners would lead to an increase of the Herfindahl index, if free float increases. This means, that the results would be biased. Excluding all companies with existing free float would also yield biased results, since companies with lower ownership concentration are systematically excluded from the sample.

One way to avoid biased results would be the use of a different concentration measure which does not require information about the size of all single owner blocks. An alternative measure that fulfills this criteria is the sum of the N largest voting blocks, where N can be randomly chosen. Unfortunately, as the study of Becht / Böhmer (1999) has shown, only a fifth of all German corporations has more than two owners. We therefore do not use alternative concentration measures in this paper. To be able to calculate the Herfindahl index for companies with free float, we split it up into single blocks and assume a size of 0.1% for each block.²⁶ Since the single blocks are squared for the calculation of the Herfindahl index, we can reduce the bias to a minimum and are still able to include companies with free float in the sample.

The main specification reads as follows:

$$\Delta H_{it} = \alpha + \beta_1 \cdot PF2_{it} + \beta_2 \cdot PF3_{it} + \beta_3 \cdot Cor_{it} + \beta_4 \cdot For_{it} + \beta_5 \cdot State_{it} + \beta \cdot X'_{it} + \varepsilon_{it} \quad (8)$$

 ΔH_{it} represents the change in ownership concentration, $PF2_{it}$ and $PF3_{it}$ as well as Cor_{it} , For_{it} and $State_{it}$ refer to the tax portfolios and owner types defined in chapter 4.1 and X'_{it} is the vector of non-tax control variables used equation 6.

Using information about the tax portfolios and owner types can be seen as a proxy for the owner's marginal tax rate on equity τ_e . For a more general estimation we calculate a firm and owner specific marginal tax rate on equity and run the following alternative

²⁶ According to this definition a free float of e.g. 5% would be split up into 50 single blocks, assuming that free float consists of 50 single individual owners.

specification:

$$\Delta H_{it} = \alpha + \beta_1 \cdot \tau_e + \beta \cdot X'_{it} + \varepsilon_{it} \tag{9}$$

As the dependant variable, we use the change in the value of the Herfindahl index for company i calculated as follows:

$$\Delta H_{it} = \Delta (H_{it} - H_{it-1}) > 0 \tag{10}$$

Additionally, we include four dummy variables referring to the four owner types defined in chapter 4.1: corporate (Cor_{it}) , individual $(Indiv_{it})$, foreign (For_{it}) and state $(State_{it})$. Since we only expect disposals to individual owners to have an influence on ownership concentration (see hypothesis 3a in chapter 3.3), we omit $Indiv_{it}$ as the reference category. Based on the results from chapter 3.1, we expect a negative coefficient (a reduction in ownership concentration) for $PF2_{it}$ and $PF3_{it}$. By contrast, we expect a positive coefficient for Cor_{it} , For_{it} and Gov_{it} .

We include two additional control variables compared equation 6: $Total_{it}$ is a dummy variable taking the value 1 if the block was bought by an existing owner, which after the transaction owns 100% (all) of the shares of the company. We expect a negative coefficient for $Total_{it}$. H_{it-1} is the absolute value of the Herfindahl index the year before the disposal and controls for different starting levels in ownership concentration.

For the alternative specification we expect a positive coefficient for τ_e , since higher marginal tax rates on equity are expected to lower the demand for blocks and therefore increase the absolute size of blocks held (see hypothesis 3b in chapter 3.3).

5 Empirical Results

In order to fully evaluate tax induced effects of TRA 2001 on changes in corporate ownership, the results are divided into two aspects (see chapter 3.2). The effects of TRA 2001 on the demand for corporate shares of single owner types is going to be analyzed in chapter 5.1, results on general ownership concentration are carried out in chapter 5.2.

5.1 Effects of TRA 2001 on the demand of single owner types

For the evaluation of the effects of TRA 2001 on the demand for corporate shares of single owner types we have divided the 459 disposals into the corresponding four owner types. The next table shows the distribution of the owner types among the three tax portfolios.

{Insert table 4 about here.}

As we can see from table 4, for all three portfolios the majority of blocks were bought by corporate owners. Still, the number of blocks bought by corporate owners is decreasing over time, resulting in lower values for portfolio 2 and 3 compared to portfolio 1. For individual owners we find the highest number of acquisitions in portfolio 2. The number of blocks bought by individual owners in portfolio 2 is more than four times the value of portfolio 1 and twice the value of portfolio 3. Foreigners or the state are only of minor importance among the owner groups observed.

Table 5 shows the distribution of single owner types among the three tax portfolios if we divide the sample according to whether the block was sold by a financial or non financial company. As we can see, disposals by financial selling companies differ substantially from disposals by non financial selling companies.

{Insert table 5 about here.}

The number of blocks acquired by individual or foreign owners is higher if the selling company belongs to the financial industry. Contrary, the number of blocks acquired by corporate owners is lower and the state is not acquiring blocks sold by financial companies at all. For portfolio 2 the majority of blocks sold by financial companies are bought by individual owners, whereas for portfolio 3 there are about as many blocks bought by individual owners as blocks bought by corporate owners. This clearly shows a short time increase of acquisitions by individual owners.

In order to get more detailed information about tax incentives for individual owners, we divide blocks bought by owner type number 2 into the three further subgroups 2a, 2b and 2c as defined in chapter 4.1.

{Insert table 6 about here.}

Taking a closer look at the subgroups of individual owners, we find individual owners holding blocks below the 1% threshold to acquire the majority of blocks in portfolio 1 and 2. As we can see from table 6, there is a sharp increase in the number of blocks bought by individual owners below the 1% threshold for portfolio 2. The number of blocks bought by individual owners below the 1% threshold for portfolio 3 is lower than for portfolio 2, but still higher than for portfolio 1. We therefore find evidence for both, a short and a long term increase in acquisitions by individual owners below the 1% threshold. The proportion of blocks acquired by individual owners above the 10% threshold is lower for portfolio 2 than for portfolio 1 and 3, indicating a short term decrease in demand. Still, comparing the values for portfolio 1 and 3, we do not find a long term change in the demand for corporate shares of individual owners above the 10% threshold. Contrary to our expectations, the number of owners below the 10% threshold and above the 1% threshold (holdings affected by the lowering of the threshold for substantial interest due to TRA 2001) is slightly increasing over time.

Table 7 shows summary statistics for the variables used in the regression analysis.

Although the sample firms distribute only about a third of their profits via dividends on average, the high standard deviation shows that there is a lot of variation in their payout policy. About a quarter of all firms are listed companies; nearly 30% of the selling companies belong to the financial industry.

Table 8 shows the results for the estimation of equation 6.

We find the probability of a corporate owner buying a block to be lower under the half income system than under the full imputation system. The coefficient for PF2 (PF3) indicates that the probability of a corporate owner buying a block under the half income system is about 13% (8%) lower than under the full imputation system. Since both coefficients are not significant, we do not find TRA 2001 to substantially lower the probability of a corporate owner buying a block. As we have expected, the dividend ratio of a firm has a positive influence on the decision to buy a block for corporate owners under the full imputation system. We find a preference for corporate owners to hold blocks in non listed companies and buy blocks from companies outside the financial sector. The probability of a corporate owner buying a block is 40% lower, if the company is listed and 26% higher if the selling company is a non financial company.

The probability of an individual owner below the 1% threshold buying a block has significantly increased due to TRA 2001 in the short run. The coefficient for PF2 is significant at the 5% level and indicates, that the probability of an individual owner below the 1% threshold buying a block under the half income system is about 12% higher than under the full imputation system. We find the expected negative influence of the company's payout policy on the probability of an individual owner below the 1% threshold buying a block for both, the full imputation and half income system. Contrary to corporate owners we find individual owners below the 1% threshold to prefer to hold blocks in listed and companies and buy block from financial companies. The probability of an individual owner if the selling company is a non financial company.

For individual owners above the 10% threshold we find the probability of buying a block to be lower under the half income system than under the full imputation system. The coefficient for PF3 is negative, but close to zero, indicating no significant change in demand due to TRA 2001 in the long run. Again, we find the expected influence of the firm's payout policy, since the coefficient of d is negative for the full imputation system.

Taken these results together, we can not find supporting evidence for hypothesis 1a and 1c, but are able to find support for hypothesis 1b, 2a, 2b, 2c and 2d.

5.2 Effects of TRA 2001 on ownership concentration

In order to test the dynamic effects of TRA 2001 on ownership concentration, we estimate equations 8 and 9.

Table 9 shows the distribution of the change in the Herfindahl index for the total sample as well as for the three tax portfolios and four owner types.

As we can see from table 9, the mean value of the change in the Herfindahl index is positive for the total sample as well as for all tax portfolios. The mean value for portfolio 1 is close to zero, showing that disposals under the full imputation system did not affect ownership concentration on average . In contrast, the positive mean value for portfolio 2 and 3 indicates that disposals under the half income system increased ownership concentration on average.

Dividing the sample into the four owner types, we see that acquisitions by individual or foreign owners reduced ownership concentration on average, whereas acquisitions by corporate owners or the state did not. For blocks acquired by the state, we also see that all acquisitions were leading to an increase in ownership concentration, since even the minimum value is positive. The fact that disposals to individual owners were reducing ownership concentration corresponds with the lowering of the threshold for substantial interest due to TRA 2001.

Table 10 shows the results for the estimation of equations 8 and 9, applying a cross section OLS regression.

{Insert table 10 about here.}

As we would have expected from summary statistics, the coefficients for PF2 and PF3 are both positive. This means, that ownership concentration is higher for portfolio 2 and 3 than for portfolio 1 and is therefore an indicator that disposals of owner blocks under the half income systems were leading to more concentrated ownership structures. We thus do not find supporting evidence for hypothesis 3a.

Looking at different owner types, we find all of the coefficients to have the expected positive sign, but only acquisitions by corporate owners and by the state to significantly differ from acquisitions by individual owners. The positive coefficients show, that according to our hypothesis acquisitions by corporate or foreign owners and the state lead to a higher increase in ownership concentration than acquisitions by individual owners.

In line with the results from chapter 5.1, the coefficient for *Listed* (NF) is negative (positive) and significant, indicating that ownership concentration is lower if the company is listed or the selling company is a financial company.

For the alternative specification, we find the coefficient for τ_e to have a significant negative influence on ownership concentration. This is not what we have expected, but corresponds with the negative coefficients of PF2 and PF3 in the basis specification. We therefore do not find supporting evidence for hypothesis 3b.

5.3 Robustness Checks

The analysis of the effects of TRA 2001 on the demand for corporate shares of single owner types has shown mixed results. In addition, we are not able to detect a tax induced decrease in ownership concentration of German corporations. We therefore apply several robustness checks to the results.

In order to check whether firm specific factors are driving the results for the demand of single owner types, we generate a vector of interactive variables by multiplying PF2 with all other non-tax control variables in the model. Non tabulated results for the extended regressions show, that results for individual owners remain unchanged. For corporate owners we find one factor to significantly drive the probability of buying the block, since the coefficient for PF2NF is positive and significant. This is again an indicator for the fact, that corporate owners prefer to acquire blocks from companies outside the financial industry. The coefficient for PF2 still shows a negative sign, indicating that even when controlling for differences with respect to financial selling companies, the probability of a corporate owner acquiring a block is lower under the half income system.

We also generate a vector of interactive variables by multiplying PF2 with all other non-tax control variables in order to control for firm specific effects regarding ownership concentration, but find none of the firm specific factors to systematically influence ownership concentration under the half income system.

One crucial assumption we have to make for the calculation of the Herfindahl index is the treatment of free float (see chapter 4.2). Results in chapter 5.2 are based on the assumption that free float is split up into single blocks of 0.1% each. Since the databases do not report single blocks below this threshold, there is no problem of underestimating ownership concentration. We could still overestimate concentration, if free float blocks can be assumed to be smaller than 0.1%. We therefore repeat the regressions assuming that free float is split up into blocks of 0.01% and 0.001% each. Results remain unchanged.

90 out of the 459 sample companies (about 20%) belong to the electricity, gas, steam and hot water supply industry. Taking a closer look at companies from this industry, we find most of them to be municipal utilities. Although municipal utilities are private companies, they can be defined as operating semi-public according to their business structure. In addition, the number and size of blocks held by the state is higher for municipal utilities. In order to test whether the results are driven by ownership changes in the semi-public industry, we exclude all municipal utilities from the sample. Results remain unchanged.

For 51 observations (about 11%) the buyer of the block is holding 100% of the shares of the company after the acquisition, resulting in an obvious increase in ownership concentration. The investment decision of owners acquiring 100% of the shares of a company might be driven by strategical business reasons rather than tax aspects. We therefore repeat the regressions excluding disposals where Total equals 1. This slightly decreases ownership

concentration for portfolio 2, but the coefficient is still positive and significant. Results for portfolio 3 and different owner groups remain unchanged.

6 Conclusion

The German TRA 2001 changed the corporate tax system from a full imputation system to a half-income system, a classical corporate tax system with shareholder relief elements. Along with the change in the tax system, the taxation of equity investments for corporate and individual investors changed as well. Whereas several papers have already analyzed the effect of TRA 2001, to the best of our knowledge, no paper has so far dealt with general ownership effects of TRA 2001.

In order to fully evaluate tax induced ownership effects, we look at two different distortional aspects. On the one hand we evaluate effects of TRA 2001 on the demand for corporate shares by analyzing whether the number of blocks bought by different owner types has changed due to TRA 2001. On the other hand we measure the influence of TRA 2001 on ownership concentration.

Calculating marginal tax rates on equity for different owner types, we show that corporate owners carry the lowest tax burden before and after the tax reform. For individual owners there is a clear tax incentive to hold shares below the 10% threshold for the full imputation system and below the 1% threshold for the half-income system. We therefore expect the number of blocks bought by corporate owners, by individual owners below the 1% threshold and by individual owners above the 10% threshold to increase due to TRA 2001. In addition, we expect the corporate dividend payout ratio to have a significant influence on the acquisition decision. Due to the lowering of the threshold for substantial interest for individual owners from 10% to 1%, we also expect a decrease in ownership concentration of German corporations.

We analyze 459 disposals of German minority blocks over the period 1997-2006 and categorize disposals into three tax portfolios according to the tax status of the seller and the buyer. Due to the definition of the three tax portfolios, we are able to distinguish between blocks where both, the acquisition and the disposal took place under the full imputation system or under the half income system. In addition, we are able to separate the effects for blocks that have been acquired under the full imputation system and sold under the half income system.

We find the probability of a corporate owner and of an individual owner above the 10% threshold buying a block to be lower under the half income system than under the full

imputation system. In addition, we find a preference for corporate owners to hold blocks in non listed companies and buy blocks from companies outside the financial sector. Contrary, the probability of an individual owner below the 1% threshold buying a block has significantly increased due to TRA 2001 in the short run. The coefficient for PF2indicates, that the probability of an individual owner below the 1% threshold buying a block under the half income system is about 12% higher than under the full imputation system. Contrary to corporate owners, we find individual owners below the 1% threshold to prefer to hold blocks in listed companies and buy blocks from financial companies. For all three owner types, we find the expected influence of the company's dividend payout ratio on the acquisition decision.

In order to test the effects of TRA 2001 on ownership concentration we run a cross section OLS regression, taking a Herfindahl index as the measure of concentration. Contrary to our expectations, we we find a significant increase in ownership concentration due to TRA 2001.

We are able to show that TRA 2001 was able to fulfill government's expectations about an increase in blocks bought by individual owners. It can also be shown, that corporate and individual owners have different preferences with respect to firm specific characteristics: corporate owners prefer to hold blocks in non listed companies and buy blocks sold by non financial companies, whereas individual owners are more likely to hold shares in listed companies and buy blocks sold by financial companies. With respect to ownership concentration, we find tax incentives not to be strong enough to lead to a reduction in overall concentration of corporate ownership.

Appendix

		Independent Variables				
Authors	Sample	Size	Control	Regulation	Amenity	Age
			Potential		Potential	
Demsetz /Lehn (1985)	511 (US)	yes; -	yes; $+$	yes; -	yes; +	no
Bergstroem / Rydquist (1990)	204 (Sweden)	yes; -	yes; $+$	yes; -	no	no
Leech / Leahy (1991)	470 (UK)	yes; -	yes; $+$	no	no	yes; n.s.
Prowse (1992)	734 (Japan)	yes; -	yes; $+$	no	no	no
Bohren / Odegaard (2001)	1255 (Norway)	yes; -	no	yes; n.s.	no	yes; n.s.
Mak / Li (2001)	147 (Singapore)	yes; n.s.	yes; n.s.	yes; n.s.	no	no
Lamba / Stapledon (2001)	240 (Australia)	yes; -	yes; n.s.	yes; -	yes; $+$	yes; n.s.
Wojcik (2003)	454 (Germany)	yes; n.s.	no	yes; -	no	yes; n.s.
Pindado / de la Torre (2003)	135 (Spain)	yes; -	no	no	no	no
Madsen et al. (2007)	1201 (Denmark)	yes; -	yes; n.s.	no	no	yes; $+$
Rogers et al. (2008)	171 (Brazil)	yes; -	yes; $+$	yes; $+$	no	no

Table 1: Variables used in empirical ownership structure research as well as results of the papers. Independent variables refer to the five most commonly used variables in research, showing whether the variable has been used in the paper (yes/no). If the variable was used, it is shown whether it had no significant influence (n.s.), a significant positive (+) or a significant negative (-) influence on ownership concentration.

d(%)	$ au_{e(cor)}^{IMP}$	$ au_{e(cor)}^{HI}$	$\tau_{e(indiv<10\%)}^{IMP}$	$\tau_{e(indiv<1\%)}^{HI}$	$\tau_{e(indiv \ge 10\%)}^{IMP}$	$\tau_{e(indiv \ge 1\%)}^{HI}$
0	0.640	0.259	0.400	0.250	0.682-0.658	0.432-0.407
0.1	0.616	0.259	0.413-0.408	0.268-0.266	0.667-0.641	0.432-0.407
0.2	0.592	0.259	0.426-0.417	0.286-0.281	0.652-0.623	0.432-0.407
0.3	0.568	0.259	0.439-0.425	0.305-0.297	0.636-0.606	0.432-0.407
0.4	0.544	0.259	0.452-0.434	0.323-0.313	0.621-0.589	0.432-0.407
0.5	0.520	0.259	0.465-0.442	0.341-0.329	0.606-0.571	0.432-0.407
0.6	0.496	0.259	0.478-0.451	0.359-0.344	0.591-0.554	0.432-0.407
0.7	0.472	0.259	0.491 - 0.459	0.377-0.360	0.576 - 0.537	0.432-0.407
0.8	0.448	0.259	0.504-0.468	0.395-0.376	0.560-0.520	0.432-0.407
0.9	0.424	0.259	0.517-0.476	0.414-0.392	0.545-0.502	0.432-0.407
1	0.400	0.259	0.530 - 0.485	0.432 - 0.407	0.530 - 0.485	0.432-0.407

Table 2: Marginal tax rate on equity (in %) for corporate and individual owners in Germany over the period 1998-2006, depending on the dividend payout ratio d. Note: The range of tax rates for individual owners is due to changes in the top statutory tax rate over time.

	PF 1	PF 2	PF 3	total
total	62	207	190	459
(%)	13.51%	45.10%	41.39%	100%
non-financial	56	125	149	330
(%)	16.97%	37.88%	45.15%	100%
financial	6	82	41	129
(%)	4.65%	63.56%	31.79%	100%

Table 3: Distribution of minority blocks among the three tax portfolios.

Owner Type	Portfolio 1	Portfolio 2	Portfolio 3
Corporate	85.48%	65.21%	80.53%
Individual	6.45%	28.51%	14.21%
Foreign	1.62%	4.34%	2.63%
State	6.45%	1.94%	2.63%
Total	100.00%	100.00%	100.00%

Table 4: Distribution of owner types among the three tax portfolios.

	Portfolio 1		Portf	olio 2	Portfolio 3	
	NF	F	NF	F	NF	F
Corporate	87.50%	66.67%	92.80%	23.17%	89.26%	48.78%
Individual	3.57%	33.33%	2.40%	68.29%	5.37%	46.34%
Foreign	1.79%	0.00%	1.60%	8.54%	2.01%	4.88%
State	7.140%	0.00%	3.20%	0.00%	3.36%	0.00%
Total	100.00%	100.00%	100.00%	100.00%	100.00%	100.00%

Table 5: Distribution of disposals by owner groups among the three portfolios for F (financial) and NF (non-financial) selling companies.

	Portfolio 1		Portfolio 2		Portfolio 3	
Individual	number	%	number	%	number	%
<1%	1	1.61%	49	23.68%	16	8.43%
$\geq 1\%$ and $<10\%$	1	1.61%	4	1.93%	7	2.10%
$\geq 10\%$	2	3.23%	6	2.90%	7	3.68%
total	4	6.45%	69	28.51%	27	14.21%

Table 6: Distribution of individual owners among the three portfolios.

Variable	Mean	Std. Dev.	Min	Max	Obs.
d	0.3384	0.4077	0	1	459
Size	21600000	115000000	19	11300000000	459
ConPot	139073.3	561430.6	0.5773	4570715	459
Age	37.8605	40.8060	1	247	459
Reg	0.2766	0.4478	0	1	459
Listed	0.2483	0.4325	0	1	459
NF	0.7189	0.4500	0	1	459
Interest	3.1248	0.8019	2.278	5.188	459
Stock	5004.361	1163.739	2930.74	6561.63	459

Table 7: Summary statistics.

	Z1 (corporate)		${f Z2a} \ ({ m individual} \le 1\%)$			${f Z2c} \ ({ m individual}>10\%)$			
	Coeff.	(Std.err)	(Marg. Eff.)	Coeff.	(Std.err)	(Marg. Eff.)	Coeff.	(Std.err)	(Marg. Eff.)
PF2(+)	-0.4483	0.3193	-0.1278	1.3976^{**}	0.6827	0.1206	-0.4627	0.5768	-0.0085
PF3(+)	-0.2720	0.3406	-0.0777	0.9404	0.7170	0.0761	-0.0561	0.5231	-0.0010
d (+/-)	0.0696	0.1452	0.0195	-0.1045	0.5326	-0.0066	0.3696	0.2458	0.0069
dPF2 (-)	-0.6911	0.1457	-0.0194	-0.1632	0.5338	-0.0104	-0.3397	0.2498	-0.0063
dPF3 (-)	-0.5361	0.1829	-0.0151	-0.1973	0.5493	-0.0126	-0.4567	0.2969	-0.0085
Size (-)	0.0000	0.0000	0.0000	-0.0000	0.0000	-0.0000	-0.0000	0.0000	-0.0000
ConPot $(+)$	-0.0000	0.0000	-0.0000	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000
Age $(+)$	0.0048***	0.0020	0.0013	-0.0011	0.0021	-0.0007	-0.0097**	0.0046	-0.0001
Reg (-)	0.5580***	0.2268	0.1399	-0.5686	0.3981	-0.2095	-0.4233	0.4948	-0.0064
Listed $(+)$	-1.211***	0.2259	-0.4013	1.3569^{***}	0.2974	0.1738	0.4601	0.3785	0.0117
NF (-)	0.8318***	0.2237	0.2632	-1.0822***	0.3229	-0.1150	-0.8073**	0.3906	-0.0255
Interest (-)	-2733	0.1720	-0.0767	-0.0471	0.2483	-0.0030	0.0401	0.2427	0.0007
Stock $(+)$	0.0002**	0.0001	0.0000	-0.2360	0.7506	-0.0151	0.4054	0.9443	0.0075
N		459			459			459	
R^2		0.3885			0.5356			0.2438	

Significance levels : *: 10% **: 5% ***: 1%

Table 8:	Estimation	results	for	equation	6.
----------	------------	---------	-----	----------	----

	Obs	Mean	Std. Dev.	Min	Max
total	459	0.0141283	0.2039373	-0.9777692	0.9677079
PF1	62	0.0008015	0.2361292	-0.9777692	0.6912500
PF2	207	0.0136548	0.2320837	-0.6607195	0.8080927
PF3	190	0.0166389	0.1551229	-0.7447016	0.9677079
Cor	341	0.0403777	0.2062282	-0.9777692	0.9677079
Indiv	90	-0.0864492	0.1695440	-0.7447016	0.4333690
For	15	-0.0090930	0.1956809	-0.4222780	0.3648000
State	13	0.1364887	0.1241351	0.0002000	0.4037488

Table 9: Summary statistics for the change in ownership concentration among portfolios and owner groups.

	est. of equation 8		est. of e	equation 9
Variable	Coeff.	(Std.Err.)	Coeff.	(Std.Err.)
PF2 (-)	0.0526**	(0.0245)		
PF3(-)	0.0339	(0.0248)		
d (-)	-0.0021^{*}	(0.0011)		
Cor(+)	0.0801^{***}	(0.0283)		
For $(+)$	0.0652	(0.0463)		
State $(+)$	0.0200^{***}	(0.0537)		
$\tau_e (+)$			-0.1128*	(0.0667)
InSize (-)	-0.0131**	(0.0055)	-0.0137**	(0.0055)
$\ln ConPot (+)$	0.0214^{***}	(0.0052)	0.0209^{***}	(0.0052)
$\operatorname{Reg}(-)$	-0.0161	(0.0180)	-0.0161	(0.0179)
Age $(+)$	0.0066^{**}	(0.0072)	0.0105	(0.0071)
Listed (-)	-0.0951^{***}	(0.0304)	-0.1204^{***}	(0.0282)
NonFinancial $(+)$	0.0495^{*}	(0.0264)	0.0683^{***}	(0.0254)
Total $(+)$	0.3220^{***}	(0.0280)	0.3224^{***}	(0.2831)
H_{it} (-)	-0.3365***	(0.0317)	-0.3287^{***}	(0.0319)
Interest $(+)$	0.0003	(0.0127)	0.0005	(0.0127)
Stock $(-)$	0.1212^{**}	(0.0512)	0.1219^{**}	(0.0497)
N		459	4	59
\mathbb{R}^2	0.	4185	0.3	3961
F (16,442)	1	9.88		
F (11,447)			26	6.66
Significance levels :	*: 10% **:	5% ***:1%		

Table 10: Estimation results for equations 8 and 9.

References

- Adams, Michael (1994): Die Ursupation von Aktionärsbefugnissen mittels Ringverflechtung in der Deutschland AG, in: Die Aktiengesellschaft 39, p. 148–158.
- Becht, Marco/ Böhmer, Ekkehard (1999): Transparency of Ownership and Control in Germany, Working Paper, http://papers.ssrn.com/sol3/papers.cfm?abstract_id=149774, 15.06.2010.
- Becht, Marco/ Röell, Ailsa (1999): Blockholdings in Europe: An International Comparison, in: European Economic Review 43, p. 1049–1056.
- Beny, Laura Nyantung (2005): Do Insider Trading Laws Matter? Some Preliminary Comparative Evidende, in: American Law and Economics Review, p. 114–183.
- Bergström, Clas/ Rydquist, Kristian (1990): The Determinants of Corporate Ownership, in: Journal of Banking and Finance 14, p. 237–253.
- Beyer, Jürgen (2003): Deutschland AG a.D.: Deutsche Bank, Allianz und das Verflechtungszentrum des deutschen Kapitalismus, in: W. Streek, M. Höpner (Ed.): Alle Macht dem Markt? Fallstudien zur Abwicklung der Deutschland AG, Frankfurt, p. 118–146.
- Bohren, Oyvind/ Odegaard, Berndt Arne (2001): Patterns of Corporate Ownership: Insights from a Unique Data Set, in: Nordic Journal of Political Economy 27, p. 55–86.
- Breuer, Rolf E. (1998): The Role of Financial Intermediaries and Capital Markets, in: K. Hopt, H. Kanda, M. Roe, E. Wymeersch, S. Prigge (Ed.): Comparative Corporate Governance - The State of the Art and Emerging Research, Oxford, p. 537–544.
- Cioffi, John W. (2002): Restructuring Germany Inc.: The Politics of Company and Takeover Law Reform in Germany and the European Union, in: Law and Policy 24, p. 355–402.
- Dahlquist, Magnus/ Robertsson, Göran (2000): Direct Foreign Ownership, Institutional Investors and Firm Characteristics, in: Journal of Finance 59, p. 413–440.
- Demsetz, Harold/ Lehn, Kenneth (1985): The Structure of Corporate Ownership: Causes and Consequences, in: Journal of Political Economy 93, p. 1155–1177.
- Desai, Mihir A./ Dharmapala, Dhammika/ Fung, Winnie (2007): Taxation and the Evolution of Aggregate Corporate Ownerhship Concentration, in: A. Auerbach,

J. Hines and J. Slemrod (Ed.): Taxing Corporate Income in the 21st Century, Cambridge, p. 345–383.

- Desai, Mihir A./ Gentry, William M. (2004): The Character and Determinants of Corporate Capital Gains, in: J. Poterba (Ed.): Tax Policy and the Economy 18, Cambrigde, p. 1–36.
- Djankov, Simeon/ La Porta, Rafael/ Lopez de Silanes, Florencio/ Shleifer, Andrei (2008): The Law and Economics of Self Dealing, in: Journal of Financial Economics 88, p. 430–465.
- Edwards, Courtney/ Lang, Mark H./ Maydew, Edward/ Shackelford, Douglas A. (2004): Germany's Repeal of the Corporate Capital Gains Tax: The Equity Market Response, in: Journal of the American Taxation Association 26, p. 73– 97.
- Frick, Bernd/ Lehmann, Erik (2004): Corporate Governance in Germany: Ownership, Codetermination and Firm Performance in a Stakeholder Economy, in: H. Gospel and A. Pendleton (Ed.): Corporate Governance and Labour Management: An International Comparision, Oxford, p. 122–147.
- Gieralka, Adam/ Drajewicz, Agnieszka (2001): Reaktion des deutschen Kapitalmarktes auf die Ankündigung und Verabschiedung der Unternehmenssteuerreform 2001, in: FinanzBetrieb 3, p. 311–321.
- Gordon, Myron J. (1963): The Investment, Financing and Valuation of the Corporation, Irwin, Homewood.
- Gordon, Myron J./ Shapiro, Eli (1956): Capital Equipment Analysis: The required Rate of Profit, in: Management Science 3, p. 102–113.
- Gordon, Roger H./ MacKie-Mason, Jeffrey K. (1990): Effects of the Tax Reform Act of 1986 on Corporate Financial Policy and Organizational Form, in: J. Slemrod (Ed.): Do Taxes Matter? The Impact of the Tax Reform Act of 1986, Cambridge, p. 50–80.
- Holderness, Clifford G. (2009):Do Differences inLegal Protection Explain Differences in Ownership Concentration?, Working Paper, http://papers.ssrn.com/sol3/papers.cfm?abstract_id=1104678, 15.06.2010.
- Jacob, Martin (2010): Why Do Some Realize Capital Gains and Others Losses? Evidence from German Income Tax Data, Working Pa-

per, http://papers.ssrn.com/sol3/papers.cfm?abstract_id=1474862, 01.10.2010.

- Kengelbach, Jens/ Roos, Alexander (2006): Entflechtung der Deutschland AG. Empirische Untersuchung der Reduktion von Kapital- und Personalverflechtungen zwischen deutschen börsennotierten Gesellschaften, in: M&A Review, p. 12–21.
- La Porta, Rafael/ Lopez de Silanes, Florencio/ Shleifer, Andre (1999): Corporate Ownership Around the World, in: Journal of Finance 54, p. 471–517.
- La Porta, Rafael/ Lopez de Silanes, Florencio/ Shleifer, Andre (2006): What Works in Security Laws?, in: Journal of Finance 61, p. 1–32.
- La Porta, Rafael/ Lopez de Silanes, Florencio/ Shleifer, Andre/ Vishny, Robert W. (1998): Law and Finance, in: Journal of Political Economy 106, p. 1113–1155.
- Lamba, Asjeet S./ Stapledon, Geof (2001): The Determinants of Corporate Ownership Structure: Australian Evidence, University of Melbourne Public Law Research Paper No. 20,http://papers.ssrn.com/sol3/papers.cfm?abstract_id=279015, 15.05.2009.
- Leech, Dennis/ Leahy, John (1991): Ownership Structure, Control Type Classifications and the Performance of Large British Companies, in: Economic Journal 101, p. 1418–1437.
- Madsen, Erik Strojer/ Smith, Valdemar/ Dilling-Hansen, Mogens (2007): Why Does Any Firm Have Several Owners?, in: Corporate Ownership and Control 4, p. 80– 86.
- Mak, Y.T./ Li, Yuan (2001): Determinants of Coporate Ownership and Board Structure: Evidence from Singapore, in: Journal of Corporate Finance 7, p. 235–256.
- Melicher, Ronald W./ Ledolter, Johannes/ D'Antonio, Louis J. (1983): A Time Series Analysis of Aggregated Merger Activity, in: Review of Economics and Statistics 65, p. 423–430.
- Pindado, Julio/ de la Torre, Chabela (2006): The Role of Investment, Financing and Dividend Decisions in Explaining Corporate Ownership Structure: Empirical Evidence from Spain, in: European Financial Management 12, p. 661–687.
- Prowse, Stephen D. (1992): The Structure of Corporate Ownership in Japan, in: Journal of Finance 47, p. 1121–1140.
- Roe, Mark J. (2006): Legal Origins, Politics and Modern Stock Market, in: Harvard Law Review 120, p. 460–527.

- Rogers, Pablo/ Borges, Anamalia/ de Sousa Ribiero, Korem Cristina/ de Sousa, Almir (2008): Corporate Governance and Ownership Structure in Brazil: Causes and Consequences, in: Corporate Ownership and Control 5, p. 36–54.
- Schürmann, Christoph (2005): Leeres Karussell, in: Wirtschaftswoche, 14.07.2005, p. 84.
- Shleifer, Andrei/ Vishny, Robert W. (1997): A Survey of Corporate Governance, in: Journal of Finance 52, p. 737–783.
- Short, Helen (1994): Ownership, Control, Financial Structure and the Performance of Firms, in: Journal of Economic Surveys 8, p. 203–249.
- Sureth, Caren/ Langeleh, Dirk (2007): The Degree of Integrating Corporate and Capital Gains Tax into Income Tax and its Impact on Investment Decisions, in: Schmalenbach Business Review 59, p. 310–339.
- Watrin, Christoph/ Benhof, Hanno (2008): Germany's Repeal of the Corporate Capital Gains Tax in 2001: Effects on the M&A Market, Working Paper Westfälische Wilhelms-Universität.
- Weber, Anke (2009): An Empirical Analysis of the 2000 Corporate Tax Reform in Germany: Effects on Ownership and Control in Listed Companies, in: International Review of Law and Economics 29, p. 57–66.
- Wieandt, Axel/ Smith, Charles (2006): Die Transformation der Deutschland AG, in: Die Bank 20, p. 14–19.
- Wojcik, Darius (2002): The Länder Are the Building Blocks of the German Capital Market, in: Regional Studies 36, p. 877–895.

Bislang erschienene arqus Diskussionsbeiträge zur Quantitativen Steuerlehre

arqus Diskussionsbeitrag Nr. 1 Rainer Niemann / Corinna Treisch: Grenzüberschreitende Investitionen nach der Steuerreform 2005 – Stärkt die Gruppenbesteuerung den Holdingstandort Österreich? *März 2005*

arqus Diskussionsbeitrag Nr. 2 Caren Sureth / Armin Voß: Investitionsbereitschaft und zeitliche Indifferenz bei Realinvestitionen unter Unsicherheit und Steuern *März 2005*

arqus Diskussionsbeitrag Nr. 3 Caren Sureth / Ralf Maiterth: Wealth Tax as Alternative Minimum Tax ? The Impact of a Wealth Tax on Business Structure and Strategy *April 2005*

arqus Diskussionsbeitrag Nr. 4 Rainer Niemann: Entscheidungswirkungen der Abschnittsbesteuerung in der internationalen Steuerplanung – Vermeidung der Doppelbesteuerung, Repatriierungspolitik, Tarifprogression – *Mai 2005*

arqus Diskussionsbeitrag Nr. 5 Deborah Knirsch: Reform der steuerlichen Gewinnermittlung durch Übergang zur Einnahmen-Überschuss-Rechnung – Wer gewinnt, wer verliert? – *August 2005*

arqus Diskussionsbeitrag Nr. 6 Caren Sureth / Dirk Langeleh: Capital Gains Taxation under Different Tax Regimes September 2005

arqus Diskussionsbeitrag Nr. 7 Ralf Maiterth: Familienpolitik und deutsches Einkommensteuerrecht – Empirische Ergebnisse und familienpolitische Schlussfolgerungen – *September 2005*

arqus Diskussionsbeitrag Nr. 8 Deborah Knirsch: Lohnt sich eine detaillierte Steuerplanung für Unternehmen? – Zur Ressourcenallokation bei der Investitionsplanung – *September 2005*

arqus Diskussionsbeitrag Nr. 9 Michael Thaut: Die Umstellung der Anlage der Heubeck-Richttafeln von Perioden- auf Generationentafeln – Wirkungen auf den Steuervorteil, auf Prognoserechnungen und auf die Kosten des Arbeitgebers einer Pensionszusage *September 2005* **arqus** Diskussionsbeitrag Nr. 10 Ralf Maiterth / Heiko Müller: Beurteilung der Verteilungswirkungen der "rot-grünen" Einkommensteuerpolitik – Eine Frage des Maßstabs – *Oktober 2005*

arqus Diskussionsbeitrag Nr. 11 Deborah Knirsch / Rainer Niemann: Die Abschaffung der österreichischen Gewerbesteuer als Vorbild für eine Reform der kommunalen Steuern in Deutschland? *November 2005*

arqus Diskussionsbeitrag Nr. 12 Heiko Müller: Eine ökonomische Analyse der Besteuerung von Beteiligungen nach dem Kirchhof'schen EStGB Dezember 2005

arqus Diskussionsbeitrag Nr. 13 Dirk Kiesewetter: Gewinnausweispolitik internationaler Konzerne bei Besteuerung nach dem Trennungs- und nach dem Einheitsprinzip *Dezember 2005*

arqus Diskussionsbeitrag Nr. 14 Kay Blaufus / Sebastian Eichfelder: Steuerliche Optimierung der betrieblichen Altersvorsorge: Zuwendungsstrategien für pauschaldotierte Unterstützungskassen *Januar 2006*

arqus Diskussionsbeitrag Nr. 15 Ralf Maiterth / Caren Sureth: Unternehmensfinanzierung, Unternehmensrechtsform und Besteuerung *Januar 2006*

arqus Diskussionsbeitrag Nr. 16 André Bauer / Deborah Knirsch / Sebastian Schanz: Besteuerung von Kapitaleinkünften – Zur relativen Vorteilhaftigkeit der Standorte Österreich, Deutschland und Schweiz – *März 2006*

arqus Diskussionsbeitrag Nr. 17 Heiko Müller: Ausmaß der steuerlichen Verlustverrechnung - Eine empirische Analyse der Aufkommens- und Verteilungswirkungen *März 2006*

arqus Diskussionsbeitrag Nr. 18 Caren Sureth / Alexander Halberstadt: Steuerliche und finanzwirtschaftliche Aspekte bei der Gestaltung von Genussrechten und stillen Beteiligungen als Mitarbeiterkapitalbeteiligungen *Juni 2006* **arqus** Diskussionsbeitrag Nr. 19

André Bauer / Deborah Knirsch / Sebastian Schanz: Zur Vorteilhaftigkeit der schweizerischen Besteuerung nach dem Aufwand bei Wegzug aus Deutschland *August 2006*

arqus Diskussionsbeitrag Nr. 20 Sebastian Schanz: Interpolationsverfahren am Beispiel der Interpolation der deutschen Einkommensteuertariffunktion 2006 *September 2006*

arqus Diskussionsbeitrag Nr. 21 Rainer Niemann: The Impact of Tax Uncertainty on Irreversible Investment *Oktober 2006*

arqus Diskussionsbeitrag Nr. 22 Jochen Hundsdoerfer / Lutz Kruschwitz / Daniela Lorenz: Investitionsbewertung bei steuerlicher Optimierung der Unterlassensalternative und der Finanzierung *Januar 2007, überarbeitet November 2007*

arqus Diskussionsbeitrag Nr. 23 Sebastian Schanz: Optimale Repatriierungspolitik. Auswirkungen von Tarifänderungen auf Repatriierungsentscheidungen bei Direktinvestitionen in Deutschland und Österreich *Januar 2007*

arqus Diskussionsbeitrag Nr. 24 Heiko Müller / Caren Sureth: Group Simulation and Income Tax Statistics - How Big is the Error? *Januar 2007*

arqus Diskussionsbeitrag Nr. 25 Jens Müller: Die Fehlbewertung durch das Stuttgarter Verfahren – eine Sensitivitätsanalyse der Werttreiber von Steuer- und Marktwerten *Februar 2007*

arqus Diskussionsbeitrag Nr. 26 Thomas Gries / Ulrich Prior / Caren Sureth: Taxation of Risky Investment and Paradoxical Investor Behavior *April 2007, überarbeitet Dezember 2007*

arqus Diskussionsbeitrag Nr. 27 Jan Thomas Martini / Rainer Niemann / Dirk Simons: Transfer pricing or formula apportionment? Taxinduced distortions of multinationals' investment and production decisions *April 2007*

arqus Diskussionsbeitrag Nr. 28 Rainer Niemann: Risikoübernahme, Arbeitsanreiz und differenzierende Besteuerung *April 2007* **arqus** Diskussionsbeitrag Nr. 29 Maik Dietrich: Investitionsentscheidungen unter Berücksichtigung der Finanzierungsbeziehungen bei Besteuerung einer multinationalen Unternehmung nach dem Einheitsprinzip *Mai 2007*

arqus Diskussionsbeitrag Nr. 30 Wiebke Broekelschen / Ralf Maiterth: Zur Forderung einer am Verkehrswert orientierten Grundstücksbewertung –Eine empirische Analyse *Mai 2007*

arqus Diskussionsbeitrag Nr. 31 Martin Weiss: How Well Does a Cash-Flow Tax on Wages Approximate an Economic Income Tax on Labor Income? *Juli 2007*

arqus Diskussionsbeitrag Nr. 32 Sebastian Schanz: Repatriierungspolitik unter Unsicherheit. Lohnt sich die Optimierung? *Oktober 2007*

arqus Diskussionsbeitrag Nr. 33

Dominik Rumpf / Dirk Kiesewetter / Maik Dietrich: Investitionsentscheidungen und die Begünstigung nicht entnommener Gewinne nach § 34a EStG November 2007, überarbeitet März 2008

arqus Diskussionsbeitrag Nr. 34 Deborah Knirsch / Rainer Niemann: Allowance for Shareholder Equity – Implementing a Neutral Corporate Income Tax in the European Union *Dezember 2007*

arqus Diskussionsbeitrag Nr. 35 Ralf Maiterth/ Heiko Müller / Wiebke Broekelschen: Anmerkungen zum typisierten Ertragsteuersatz des IDW in der objektivierten Unternehmensbewertung *Dezember 2007*

arqus Diskussionsbeitrag Nr. 36 Timm Bönke / Sebastian Eichfelder: Horizontale Gleichheit im Abgaben-Transfersystem: Eine Analyse äquivalenter Einkommen von Arbeitnehmern in Deutschland *Januar 2008*

arqus Diskussionsbeitrag Nr. 37 Deborah Knirsch / Sebastian Schanz: Steuerreformen durch Tarif- oder Zeiteffekte? Eine Analyse am Beispiel der Thesaurierungsbegünstigung für Personengesellschaften *Januar 2008* arqus Diskussionsbeitrag Nr. 38

Frank Hechtner / Jochen Hundsdoerfer: Die missverständliche Änderung der Gewerbesteueranrechnung nach § 35 EStG durch das Jahressteuergesetz 2008 – Auswirkungen für die Steuerpflichtigen und für das Steueraufkommen *Februar 2008*

arqus Diskussionsbeitrag Nr. 39 Alexandra Maßbaum / Caren Sureth: The Impact of Thin Capitalization Rules on Shareholder Financing *Februar 2008*

arqus Diskussionsbeitrag Nr. 40

Rainer Niemann / Christoph Kastner: Wie streitanfällig ist das österreichische Steuerrecht? Eine empirische Untersuchung der Urteile des österreichischen Verwaltungsgerichtshofs nach Bemessungsgrundlagen-, Zeit- und Tarifeffekten *Februar 2008*

arqus Diskussionsbeitrag Nr. 41

Robert Kainz / Deborah Knirsch / Sebastian Schanz: Schafft die deutsche oder österreichische Begünstigung für thesaurierte Gewinne höhere Investitionsanreize? *März* 2008

arqus Diskussionsbeitrag Nr. 42 Henriette Houben / Ralf Maiterth: Zur Diskussion der Thesaurierungsbegünstigung nach § 34a EStG *März* 2008

arqus Diskussionsbeitrag Nr. 43

Maik Dietrich / Kristin Schönemann: Steueroptimierte Vermögensbildung mit Riester-Rente und Zwischenentnahmemodell unter Berücksichtigung der Steuerreform 2008/2009 *März* 2008

arqus Diskussionsbeitrag Nr. 44 Nadja Dwenger: Tax loss offset restrictions – Last resort for the treasury? An empirical evaluation of tax loss offset restrictions based on micro data. *Mai 2008*

arqus Diskussionsbeitrag Nr. 45

Kristin Schönemann / Maik Dietrich: Eigenheimrentenmodell oder Zwischenentnahmemodell – Welche Rechtslage integriert die eigengenutzte Immobilie besser in die Altersvorsorge? *Juni 2008*

arqus Diskussionsbeitrag Nr. 46 Christoph Sommer: Theorie der Besteuerung nach Formula Apportionment – Untersuchung auftretender ökonomischer Effekte anhand eines Allgemeinen Gleichgewichtsmodells *Juli 2008* arqus Diskussionsbeitrag Nr. 47

André Bauer / Deborah Knirsch / Rainer Niemann / Sebastian Schanz: Auswirkungen der deutschen Unternehmensteuerreform 2008 und der österreichischen Gruppenbesteuerung auf den grenzüberschreitenden Unternehmenserwerb *Juli 2008*

arqus Diskussionsbeitrag Nr. 48 Dominik Rumpf: Zinsbereinigung des Eigenkapitals im internationalen Steuerwettbewerb – Eine kostengünstige Alternative zu "Thin Capitalization Rules"? *August 2008*

arqus Diskussionsbeitrag Nr. 49 Martin Jacob: Welche privaten Veräußerungsgewinne sollten besteuert werden? *August 2008*

arqus Diskussionsbeitrag Nr. 50

Rebekka Kager/ Deborah Knirsch/ Rainer Niemann: Steuerliche Wertansätze als zusätzliche Information für unternehmerische Entscheidungen? – Eine Auswertung von IFRS-Abschlüssen der deutschen DAX-30- und der österreichischen ATX-Unternehmen – *August 2008*

arqus Diskussionsbeitrag Nr. 51

Rainer Niemann / Caren Sureth: Steuern und Risiko als substitutionale oder komplementäre Determinanten unternehmerischer Investitionspolitik? – Are taxes and risk substitutional or complementary determinants of entrepreneurial investment policy?

August 2008

arqus Diskussionsbeitrag Nr. 52

Frank Hechtner / Jochen Hundsdoerfer: Steuerbelastung privater Kapitaleinkünfte nach Einführung der Abgeltungsteuer unter besonderer Berücksichtigung der Günstigerprüfung: Unsystematische Grenzbelastungen und neue Gestaltungsmöglichkeiten *August 2008*

arqus Diskussionsbeitrag Nr. 53

Tobias Pick / Deborah Knirsch / Rainer Niemann: Substitutions- oder Komplementenhypothese im Rahmen der Ausschüttungspolitik schweizerischer Kapitalgesellschaften – eine empirische Studie *August 2008*

arqus Diskussionsbeitrag Nr. 54 Caren Sureth / Michaela Üffing: Proposals for a European Corporate Taxation and their Influence on Multinationals' Tax Planning *September 2008* arqus Diskussionsbeitrag Nr. 55 Claudia Dahle / Caren Sureth: Income-related minimum taxation concepts and their impact on corporate investment decisions *Oktober 2008*

arqus Diskussionsbeitrag Nr. 56 Dennis Bischoff / Alexander Halberstadt / Caren Sureth: Internationalisierung, Unternehmensgröße und Konzernsteuerquote *Oktober 2008*

arqus Diskussionsbeitrag Nr. 57

Nadja Dwenger / Viktor Steiner: Effective profit taxation and the elasticity of the corporate income tax base – Evidence from German corporate tax return data *November 2008*

arqus Diskussionsbeitrag Nr. 58 Martin Jacob / Rainer Niemann / Martin Weiß: The Rich Demystified – A Reply to Bach, Corneo, and Steiner (2008) *November 2008*

arqus Diskussionsbeitrag Nr. 59 Martin Fochmann / Dominik Rumpf: – Modellierung von Aktienanlagen bei laufenden Umschichtungen und einer Besteuerung von Veräußerungsgewinnen *Dezember 2008*

arqus Diskussionsbeitrag Nr. 60 Corinna Treisch / Silvia Jordan: Eine Frage der Perspektive? – Die Wahrnehmung von Steuern bei Anlageentscheidungen zur privaten Altersvorsorge *Dezember 2008*

arqus Diskussionsbeitrag Nr. 61 Nadja Dwenger / Viktor Steiner: Financial leverage and corporate taxation Evidence from German corporate tax return data *Februar 2009*

arqus Diskussionsbeitrag Nr. 62 Ute Beckmann / Sebastian Schanz: Investitions- und Finanzierungsentscheidungen in Personenunternehmen nach der Unternehmensteuerreform 2008 *Februar 2009*

arqus Diskussionsbeitrag Nr. 63 Sebastian Schanz/ Deborah Schanz: Die erbschaftsteuerliche Behandlung wiederkehrender Nutzungen und Leistungen – Zur Vorteilhaftigkeit des § 23 ErbStG *März* 2009

arqus Diskussionsbeitrag Nr. 64 Maik Dietrich: Wie beeinflussen Steuern und Kosten die Entscheidungen zwischen direkter Aktienanlage und Aktienfondsinvestment? *März 2009* **arqus** Diskussionsbeitrag Nr. 65 Maik Dietrich / Kristin Schönemann: Unternehmensnachfolgeplanung innerhalb der Familie: Schenkung oder Kauf eines Einzelunternehmens nach der Erbschaftsteuerreform? *März 2009*

arqus Diskussionsbeitrag Nr. 66 Claudia Dahle / Michaela Bäumer: Cross-Border Group-Taxation and Loss-Offset in the EU - An Analysis for CCCTB (Common Consolidated Corporate Tax Base) and ETAS (European Tax Allocation System) -*April 2009*

arqus Diskussionsbeitrag Nr. 67 Kay Blaufus / Jochen Hundsdoerfer / Renate Ortlieb: Non scholae, sed fisco discimus? Ein Experiment zum Einfluss der Steuervereinfachung auf die Nachfrage nach Steuerberatung *Mai 2009*

arqus Diskussionsbeitrag Nr. 68 Hans Dirrigl: Unternehmensbewertung für Zwecke der Steuerbemessung im Spannungsfeld von Individualisierung und Kapitalmarkttheorie – Ein aktuelles Problem vor dem Hintergrund der Erbschaftsteuerreform *Mai 2009*

arqus Diskussionsbeitrag Nr. 69 Henriette Houben / Ralf Maiterth: Zurück zum Zehnten: Modelle für die nächste Erbschaftsteuerreform *Mai 2009*

arqus Diskussionsbeitrag Nr. 70 Christoph Kaserer / Leonhard Knoll: Objektivierte Unternehmensbewertung und Anteilseignersteuern *Mai 2009*

arqus Diskussionsbeitrag Nr. 71 Dirk Kiesewetter / Dominik Rumpf: Was kostet eine finanzierungsneutrale Besteuerung von Kapitalgesellschaften? *Mai 2009*

arqus Diskussionsbeitrag Nr. 72 Rolf König: Eine mikroökonomische Analyse der Effizienzwirkungen der Pendlerpauschale *Mai 2009*

arqus Diskussionsbeitrag Nr. 73 Lutz Kruschwitz / Andreas Löffler: Do Taxes Matter in the CAPM? *Mai 2009* **arqus** Diskussionsbeitrag Nr. 74 Hans-Ulrich Küpper: Hochschulen im Umbruch *Mai 2009*

arqus Diskussionsbeitrag Nr. 75 Branka Lončarević / Rainer Niemann / Peter Schmidt: Die kroatische Mehrwertsteuer – ursprüngliche Intention, legislative und administrative Fehlentwicklungen *Mai 2009*

arqus Diskussionsbeitrag Nr. 76 Heiko Müller / Sebastian Wiese: Ökonomische Wirkungen der Missbrauchsbesteuerung bei Anteilsveräußerung nach Sacheinlage in eine Kapitalgesellschaft *Mai 2009*

arqus Diskussionsbeitrag Nr. 77 Rainer Niemann / Caren Sureth: Investment effects of capital gains taxation under simultaneous investment and abandonment flexibility *Mai 2009*

arqus Diskussionsbeitrag Nr. 78 Deborah Schanz / Sebastian Schanz: Zur Unmaßgeblichkeit der Maßgeblichkeit – Divergieren oder konvergieren Handels- und Steuerbilanz? *Mai 2009*

arqus Diskussionsbeitrag Nr. 79 Jochen Sigloch: Ertragsteuerparadoxa – Ursachen und Erklärungsansätze *Mai 2009*

arqus Diskussionsbeitrag Nr. 80

Hannes Streim / Marcus Bieker: Verschärfte Anforderungen für eine Aktivierung von Kaufpreisdifferenzen – Vorschlag zur Weiterentwicklung der Rechnungslegung vor dem Hintergrund jüngerer Erkenntnisse der normativen und empirischen Accounting-Forschung *Mai 2009*

arqus Diskussionsbeitrag Nr. 81 Ekkehard Wenger: Muss der Finanzsektor stärker reguliert werden? *Mai 2009*

arqus Diskussionsbeitrag Nr. 82 Magdalene Gruber / Nicole Höhenberger / Silke Höserle / Rainer Niemann: Familienbesteuerung in Österreich und Deutschland – Eine vergleichende Analyse unter Berücksichtigung aktueller Steuerreformen *Juni 2009*

arqus Diskussionsbeitrag Nr. 83 Andreas Pasedag: Paradoxe Wirkungen der Zinsschranke *Juli 2009* **arqus** Diskussionsbeitrag Nr. 84 Sebastian Eichfelder: Bürokratiekosten der Besteuerung: Eine Auswertung der empirischen Literatur *Juli 2009*

arqus Diskussionsbeitrag Nr. 85 Wiebke Broekelschen / Ralf Maiterth: Gleichmäßige Bewertung von Mietwohngrundstücken durch das neue steuerliche Ertragswertverfahren? Eine empirische Analyse September 2009

arqus Diskussionsbeitrag Nr. 86

Ute Beckmann / Sebastian Schanz: Optimale Komplexität von Entscheidungsmodellen unter Berücksichtigung der Besteuerung – Eine Analyse im Fall der Betriebsveräußerung September 2009

arqus Diskussionsbeitrag Nr. 87

Wiebke Breokelschen/ Ralf Maiterth: Verfassungskonforme Bewertung von Ein- und Zweifamilienhäusern nach der Erbschaftsteuerreform 2009?– Eine empirische Analyse September 2009

arqus Diskussionsbeitrag Nr. 88 Martin Weiss: How Do Germans React to the Commuting Allowance? *October 2009*

arqus Diskussionsbeitrag Nr. 89 Tobias Pick / Deborah Schanz / Rainer Niemann: Stock Price Reactions to Share Repurchase Announcements in Germany – Evidence from a Tax Perspective *October 2009*

arqus Diskussionsbeitrag Nr. 90 Wiekbe Broeckelschen: Welche Faktoren beeinflussen die Gleichmäßigkeit der Bewertung von Mietwohngrundstücken? *November 2009*

arqus Diskussionsbeitrag Nr. 91 Caren Sureth / Pia Vollert: Verschärfung der Verlustabzugsbeschränkung durch § 8c KStG und deren Einfluss auf den Erwerb von Anteilen an Kapitalgesellschaften November 2009

arqus Diskussionsbeitrag Nr. 92 Martin Fochmann / Dirk Kiesewetter / Abdolkarim Sadrieh: The Perception of Income Taxation on Risky Investments – an experimental analysis of different methods of loss Compensation – *November 2009* **arqus** Diskussionsbeitrag Nr. 93 Nadja Dwenger: Corporate taxation and investment: Explaining investment dynamics with form-level panel data *Dezember 2009*

arqus Diskussionsbeitrag Nr. 94 Kristin Schönemann: Finanzierungsstrategien und ihre Auswirkungen auf den Unternehmenswert deutscher Immobilien-Kapitalgesellschaften *Dezember 2009*

arqus Diskussionsbeitrag Nr. 95 Henriette Houben / Ralf Maiterth: Inheritance tax-exempt transfer of German businesses: Imperative or unjustified subsidy? – An empirical analysis *Dezember 2009*

arqus Diskussionsbeitrag Nr. 96 Markus Diller / Andreas Löffler: Erbschaftsteuer und Unternehmensbewertung *Februar 2010*

arqus Diskussionsbeitrag Nr. 97 Georg Schneider / Caren Sureth: The Impact of Profit Taxation on Capitalized Investment with Options to Delay and Divest *Februar 2010*

arqus Diskussionsbeitrag Nr. 98 Andreas Löffler / Lutz Kruschwitz: Ist Steuerminimierung irrational? *Februar 2010*

arqus Diskussionsbeitrag Nr. 99 Martin Fochmann / Dirk Kiesewetter / Kay Blaufus / Jochen Hundsdoerfer / Joachim Weimann: Tax Perception – an empirical survey *März* 2010

arqus Diskussionsbeitrag Nr. 100 Tasja Klotzkowski / Alexandra Maßbaum / Caren Sureth: Zinsabzugsbeschränkung durch die Zinsschranke, Fremdkapitalsteuerschild und unternehmerische Kapitalstrukturentscheidungen *April 2010*

arqus Diskussionsbeitrag Nr. 101 Frank Hechtner / Jochen Hundsdoerfer / Christian Sielaff: Zur Bedeutung von Progressionseffekten für die Steuerplanung – Eine Analyse am Beispiel der Thesaurierungsbegünstigung *April 2010*

arqus Diskussionsbeitrag Nr. 102 Henriette Houben / Ralf Maiterth: ErbSiHM 0.1 *April 2010* **arqus** Diskussionsbeitrag Nr. 103 Ralf Ewert / Rainer Niemann: Haftungsbeschränkungen, asymmetrische Besteuerung und die Bereitschaft zur Risikoübernahme – Weshalb eine rechtsformneutrale Besteuerung allokativ schädlich ist *Mai 2010*

arqus Diskussionsbeitrag Nr. 104 Frank Hechtner: Zur Bedeutung von Grenzsteuersätzen bei der Beurteilung von Tarifverwerfungen – Eine theoretische und empirische Analyse am Beispiel von § 32b EStG und § 34 EStG *Mai 2010*

arqus Diskussionsbeitrag Nr. 105 Henriette Houben / Ralf Maiterth / Heiko Müller: Aufkommens- und Verteilungsfolgen des Ersatzes des deutschen einkommensteuerlichen Formeltarifs durch einen Stufentarif *Juni 2010*

arqus Diskussionsbeitrag Nr. 106 Kay Blaufus / Jonathan Bob / Jochen Hundsdoefer / Dirk Kiesewetter / Joachim Weimann: It's All About Tax Rates – An Empirical Study of Tax Perception *November 2009*

arqus Diskussionsbeitrag Nr. 107 Lutz Kruschwitz / Andreas Löffler / Waldemar von Lehna: Was tun? *Juli 2010*

arqus Diskussionsbeitrag Nr. 108 Jens Müller / Caren Sureth: Empirische Analyse der Unternehmensbewertung für die Erbschaftsteuer mit dem vereinfachten Ertragswertverfahren *Juli 2010*

arqus Diskussionsbeitrag Nr. 109 Magdalena Haring / Rainer Niemann: Corporate Financial Policy and Investor Taxation in Austria – an Empirical Investigation – *Oktober 2010*

arqus Diskussionsbeitrag Nr. 110 Rainer Niemann: Zum Einfluß asymmetrischer Besteuerung auf die Vorteilhaftigkeit erfolgsabhängiger Entlohnungsverträge *Dezember 2010*

arqus Diskussionsbeitrag Nr. 111 Martina Corsten / Dirk Simons / Dennis Voeller: Ökonomische Anreize zur Nutzung erbschaftsteuerlicher Verschonungsregeln für das Betriebsvermögen *Dezember 2010* **arqus** Diskussionsbeitrag Nr. 112 Thi Phuong Hoa Nguyen / Sebastian Schanz: Zur Vorteilhaftigkeit von Photovoltaikanlagen unter Berücksichtigung der Besteuerung *Dezember 2010*

arqus Diskussionsbeitrag Nr. 113 Sven Arnold / Alexander Lahmann / Berhard Schwetzler: Tax Shield, Insolvenz und Zinsschranke *Januar 2011*

arqus Diskussionsbeitrag Nr. 114 Silke Rünger: The Effect of Germany's Tax Reform Act 2001 on Corporate Ownership – Insights from Disposals of Minority Blocks *Januar 2011*

Impressum:

Arbeitskreis Quantitative Steuerlehre, arqus, e.V. Vorstand: Prof. Dr. Jochen Hundsdoerfer, Prof. Dr. Dirk Kiesewetter, Prof. Dr. Ralf Maiterth Sitz des Vereins: Berlin

Herausgeber: Kay Blaufus, Jochen Hundsdoerfer, Dirk Kiesewetter, Rolf J. König, Lutz Kruschwitz, Andreas Löffler, Ralf Maiterth, Heiko Müller, Rainer Niemann, Deborah Schanz, Caren Sureth, Corinna Treisch

Kontaktadresse: Prof. Dr. Caren Sureth, Universität Paderborn, Fakultät für Wirtschaftswissenschaften, Warburger Str. 100, 33098 Paderborn, www.arqus.info, Email: info@arqus.info

ISSN 1861-8944